Author's response to reviews

Title: Tuberculous dilated cardiomyopathy: an under-recognized entity?

Authors:

Ritesh Agarwal (riteshpqi@gmail.com)
Puneet Malhotra (dranshpuneet@yahoo.co.in)
Anshu Awasthi (drshupuneet@yahoo.co.in)
Nandita Kakkar (riteshpqi@gmail.com)
Dheeraj Gupta (dheeraj@indiachest.org)

Version: 3 Date: 25 April 2005

Author's response to reviews: see over
From,
Dr. Ritesh Agarwal
Assistant Professor,
Department of Pulmonary Medicine
Postgraduate Institute of Medical Education and Research
Chandigarh 160012 (India)
Phone: (0172) 2784976
Fax: (0172) 2745959
E-mail: riteshpgi@gmail.com, drritesh1@rediffmail.com

To,
The Editor,
BMC Infectious Diseases.

Sir/Madam,
Sub: Submission of final version manuscript MS: 2040585737629388 - Tuberculous dilated cardiomyopathy: an under-recognized entity?

Herewith submitting the final version of the manuscript titled “Tuberculous dilated cardiomyopathy: an under-recognized entity?” for publication in the “Case report” section of the journal. All the corrections asked for by the editorial board have been incorporated. Hope you find the manuscript suitable for publication.

Thanking You,
Kind Regards,
Yours sincerely,
Dr. Ritesh Agarwal